Author's response to reviews

Title: Strategies for effective goals of care discussions and decision-making: Perspectives from a multi-centre survey of Canadian hospital-based healthcare providers

Authors:

- Amanda L Roze des Ordons (amanda.rozedesordons@ucalgary.ca)
- Nishan Sharma (sharmn@ucalgary.ca)
- Daren K Heyland (dkh2@queensu.ca)
- John J You (jyou@mcmaster.ca)

Version: 2 Date: 10 December 2014

Author's response to reviews:

1. Line and page numbering was added to the manuscript.
2. We have added in the specification that consent was implied by return of a completed survey (lines 129-130).